Author’s response to reviews

Title: HealthKick: a nutrition and physical activity intervention for primary schools in low-income settings

Authors:

Catherine E Draper (catherine.draper@uct.ac.za)
Anniza de Villiers (anniza.de.villiers@mrc.ac.za)
Estelle V Lambert (vicki.lambert@uct.ac.za)
Jean Fourie (jean.fourie@mrc.ac.za)
Jillian Hill (jillian.hill@mrc.ac.za)
Lucinda Dalais (ldalais@hsrc.ac.za)
Zulfa Abrahams (zabrahams@hsrc.ac.za)
Nelia P Steyn (npsteyn@hsrc.ac.za)

Version: 2 Date: 2 July 2010

Author’s response to reviews:

To the editor

Thank you again for considering this manuscript for publication in BMC Public Health, and we appreciate your patience in receiving this revised manuscript.

The revised manuscript addresses the following issues, as requested by the Editorial team:

- Consent: This has been addressed at the end of the 'Methods/Design' in the revised manuscript (paragraph 2, page 9).

- Power calculation: A comment has been inserted under the section on 'Formative assessment' (paragraph 1, page 5) which describes the sample size calculation for the situational analysis. A paragraph has been inserted under the section on 'Outcome evaluation' to describe the sample size calculation for this component of the study (paragraph 7, page 8).

We trust that these additions to the manuscript shall meet with your approval. All changes and additions to the manuscript have been highlighted with 'track changes'. Should any further information be required, please do not hesitate to contact us.

Yours sincerely

Catherine Draper